Author's response to reviews

Title: Case Report: A Rare Lesion Around The Knee Synovial Hemangiohamartomas

Authors:

Serkan Senol (drserkansenol@yahoo.com)
Hakan Cift (hakanturancift@yahoo.com)
Korhan Ozkan (korhanozkan@hotmail.com)
Esat Uygur (esatuygur@hotmail.com)
Maria Silvia Spinelli (msilvia@yahoo.it)

Version: 5 Date: 26 April 2012

Author's response to reviews:

We tried to make the necessary corrections as the reviewer’s mentioned.

Answers to Reviewer 1

Many spelling and grammatical errors were corrected. We tried to make the necessary corrections as you mentioned.

We changed the sentence starting with ‘We aimed to present a case of…..’ as ‘We present a case of....

In the abstract a number of errors with full stops were changed:
‘caucasian’ had been changed as capital ‘C’.
‘Spontaneous’ and ‘nerve’and ‘teenagers’ words were corrected.
‘Just beneath the patella’ was changed as “posterior to the patellar tendon”.

Declaration of competing interests:: I declare that we have no competing interests.

Answers to Reviewer 2

In the abstract part grammar, punctuation and syntax errors were revised.

“This lesion although rare should be thought in differential diagnosis” was changed as “This lesion, although rare should be considered as a differential diagnosis”

“pain and swelling was especially aggravated” were changed to “pain and swelling were especially aggravated”

“peripheral nerve tissue”, “teenagers”, “synovial”, discrepancy and “longitudinal” were corrected in two occasions in the abstract and main body.

The actual findings of the biopsy were clarified, especially the size of the specimen.
We decided to do open surgery because of the unknown characteristic of the lesion. The lesion which measured 3x2 cm dimension was excised fully. Some language corrections were done.

Declaration of competing interests: I declare that we have no competing interests.

Answers to Reviewer 3:
grammatical corrections were done.